



Kim, Jessica S; Oldenburg, Catherine E; Cooley, Gretchen; Amza,
Abdou; Kadri, Boubacar; Nassirou, Baido; Cotter, Sun Yu; Stoller,
Nicole E; West, Sheila K; Bailey, Robin L; Keenan, Jeremy D; Gaynor,
Bruce D; Porco, Travis C; Lietman, Thomas M; Martin, Diana L
(2019) Community-level chlamydial serology for assessing trachoma
elimination in trachoma-endemic Niger. PLoS neglected tropical diseases, 13 (1). e0007127. ISSN 1935-2727 DOI: https://doi.org/10.1371/journal.pntd.0007127

Downloaded from: http://researchonline.lshtm.ac.uk/4652171/

DOI: 10.1371/journal.pntd.0007127

Usage Guidelines

 $Please \ refer \ to \ usage \ guidelines \ at \ http://researchonline.lshtm.ac.uk/policies.html \ or \ alternatively \ contact \ researchonline@lshtm.ac.uk.$

Available under license: http://creativecommons.org/licenses/by/2.5/



Citation: Kim JS, Oldenburg CE, Cooley G, Amza A, Kadri B, Nassirou B, et al. (2019) Community-level chlamydial serology for assessing trachoma elimination in trachoma-endemic Niger. PLoS Negl Trop Dis 13(1): e0007127. https://doi.org/10.1371/ journal.pntd.0007127

Editor: Abiola Senok, Mohammed Bin Rashid University of Medicine and Health Sciences, UNITED ARAB EMIRATES

Received: October 18, 2018

Accepted: January 4, 2019

Published: January 28, 2019

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: All data underlying these analyses can be found at https://doi.org/10.7910/DVN/BRCBFA.

Funding: This work was supported by the Bill and Melinda Gates Foundation (Grant 48027; PI West), Research to Prevent Blindness, That Man May See, and the Harper-Ingles Trust. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

Community-level chlamydial serology for assessing trachoma elimination in trachomaendemic Niger

Jessica S. Kim¹, Catherine E. Oldenburg^{1,2,3}, Gretchen Cooley⁴, Abdou Amza⁵, Boubacar Kadri⁵, Baido Nassirou⁵, Sun Yu Cotter¹, Nicole E. Stoller¹, Sheila K. West⁶, Robin L. Bailey⁷, Jeremy D. Keenan^{1,2,3}, Bruce D. Gaynor^{1,2}, Travis C. Porco^{1,2,3,8}, Thomas M. Lietman^{1,2,3,8}, Diana L. Martin⁹

1 F. I. Proctor Foundation, University of California San Francisco, San Francisco, California, United States of America, 2 Department of Ophthalmology, University of California San Francisco, San Francisco, California, United States of America, 3 Department of Epidemiology and Biostatistics, University of California San Francisco, San Francisco, California, United States of America, 4 Divison of Parasitic Diseases and Malaria, Centers for Disease Prevention and Control, Atlanta, Georgia, United States of America, 5 Programme FSS/ Université Abdou Moumouni de Niamey, Programme Nationale des Soins Oculaire, Niger, 6 Dana Center for Preventive Ophthalmology, Wilmer Eye Institute, Johns Hopkins University, Baltimore, Maryland, United States of America, 7 Clinical Research Unit, Department of Infectious and Tropical Diseases, London School of Hygiene and Tropical Medicine, London, United Kingdom, 8 Institute for Global Health, University of California San Francisco, San Francisco, San Francisco, California, United States of America

* hzx3@cdc.gov

Abstract

Background

Program decision-making for trachoma elimination currently relies on conjunctival clinical signs. Antibody tests may provide additional information on the epidemiology of trachoma, particularly in regions where it is disappearing or elimination targets have been met.

Methods

A cluster-randomized trial of mass azithromycin distribution strategies for trachoma elimination was conducted over three years in a mesoendemic region of Niger. Dried blood spots were collected from a random sample of children aged 1–5 years in each of 24 study communities at 36 months after initiation of the intervention. A multiplex bead assay was used to test for antibodies to two *Chlamydia trachomatis* antigens, Pgp3 and CT694. We compared seropositivity to either antigen to clinical signs of active trachoma (trachomatous inflammation—follicular [TF] and trachomatous inflammation—intense [TI]) at the individual and cluster level, and to ocular chlamydia prevalence at the community level.

Results

Of 988 children with antibody data, TF prevalence was 7.8% (95% CI 6.1 to 9.5) and TI prevalence was 1.6% (95% CI 0.9 to 2.6). The overall prevalence of antibody positivity to Pgp3 was 27.2% (95% CI 24.5 to 30), and to CT694 was 23.7% (95% CI 21 to 26.2). Ocular chlamydia infection prevalence was 5.2% (95% CI 2.8 to 7.6). Seropositivity to Pgp3 and/or

Competing interests: The authors have declared that no competing interests exist.

CT694 was significantly associated with TF at the individual and community level and with ocular chlamydia infection and TI at the community level. Older children were more likely to be seropositive than younger children.

Conclusion

Seropositivity to Pgp3 and CT694 correlates with clinical signs and ocular chlamydia infection in a mesoendemic region of Niger.

Trial registration

ClinicalTrials.gov NCT00792922.

Author summary

Trachoma programs currently use the clinical sign of trachomatous inflammation-follicular (TF) to guide community treatment decisions and evaluate response to mass drug administration with azithromycin. These programs rely on clinical grading that poorly correlates with infection with the causative agent of trachoma, *Chlamydia trachomatis* (Ct), in low prevalence areas. Serologic measures of Ct may provide additional information about exposure and transmission patterns. Here, we evaluated the relationship between serologic markers of Ct, infection, and TF at the individual and community levels to evaluate the utility of serology for measuring trachoma in a mesoendemic region of Niger. We found that serologic markers correlated with both infection and TF, indicating that inclusion of serologic markers may be useful to guide trachoma decision making.

Introduction

Trachoma, caused by repeated ocular infection with *Chlamydia trachomatis* (*Ct*), has been targeted by the World Health Organization (WHO) for elimination as a public health problem by the year 2020. As part of the strategy to achieve elimination, WHO recommends annual mass drug administration (MDA) of azithromycin in endemic districts [1]. Program targets related to MDA focus on the district-level prevalence of trachomatous inflammation—follicular (TF) amongst children aged 1–9 years. To monitor progress towards elimination, population-based impact surveys are recommended to evaluate whether a district has reached the threshold of less than 5% TF prevalence in 1–9-year-olds and can cease azithromycin distribution. Two years after cessation of MDA, a surveillance survey to ensure that district-wide TF prevalence in 1–9-year-olds remains below 5% is conducted prior to the validation process. Currently, there are no guidelines for post-validation surveillance.

These surveys rely on a clinical grading scheme that is relatively inexpensive and simple to perform, but is poorly correlated with ocular *Ct* infection in low-prevalence settings [2]. Following MDA, the clinical sign trachomatous inflammation—intense (TI) has been shown to correlate better with infection than TF does [3]. However the measurement of clinical signs is subject to inter-grader variability and lack of real-time auditing since grading is performed in the field and thus can only later be validated or audited if images are taken. As trachoma elimination programs stand to benefit from an accurate, reproducible assessment of trachoma prevalence, other testing methods may be useful to help guide program decisions. These include

tests of infection (polymerase chain reaction [PCR] testing of ocular swabs) and antibodybased testing [4–7].

Antibodies to *Ct* antigens may act as markers of cumulative exposure to *Ct*. Two previously described *Ct* antigens, Pgp3 and CT694, have been shown to be reactive against sera in young children living in trachoma-endemic communities [4,7,8]. At the individual level, antibodies to these proteins demonstrate high sensitivity to ocular infection and high specificity against non-endemic control specimens [8–10]. However, individual associations may not always hold at the community level, and trachoma elimination programs treat ocular *Ct* infection on a population level. Additionally, as antibody markers are not yet widely used to assess for *Ct* prevalence, better characterization of how seropositivity compares to other methods of assessing trachoma prevalence is necessary. Here, we evaluate the association between seropositivity, PCR positivity, and clinical signs of active trachoma (TF and TI) at the individual and community level in a region of Niger where some trachoma transmission is occurring (TF prevalence approximately 25% at baseline). Data were collected during the final follow-up visit of the Partnership for the Rapid Elimination of Trachoma (PRET)-Niger trial, in which communities were randomized to receive annual or biannual oral azithromycin for 3 years in order to assess the impact of treatment frequency on ocular chlamydia infection [11].

Methods

Study design

The study methods have been previously reported in detail elsewhere [11–13]. Briefly, a cluster randomized trial of annual versus biannual mass azithromycin distribution for trachoma control was conducted in the Matameye district of the Zinder region of Niger from May 2010 until August 2013 [4–6]. Data on active trachoma and ocular infection were collected biannually on children aged 0–5 years; dried blood spots for serological analysis were collected only at the 36-month time point and only from children aged 1–5 years. Dried blood spots were shipped to CDC at ambient temperature and tested for antibodies from July to August 2014.

Site selection

Communities were chosen from among six different catchment areas for primary health care facilities and were eligible for inclusion if they met the following criteria: (1) contained a population between 250 to 600 persons, (2) were located more than 4 kilometers from the center of any semi-urban area, and (3) had a prevalence of active trachoma more than 10% in children aged 0–5 years [11]. 235 communities in the 6 health centers were deemed eligible, of which 48 were randomly selected for inclusion in the trial. Children aged 1–5 years were included in this analysis, due to the inability of antibody tests to differentiate between maternal-child antibodies in <1-year-olds.

Community randomization

48 communities were randomly divided into 4 treatment arms in a 2x2 factorial design (12 communities per arm), comparing two azithromycin coverage targets (standard versus enhanced coverage) and annual versus biannual treatment. Randomization of communities to treatment arms was done using RANDOM and SORT functions in Microsoft Excel (Version 2003). Only communities from the enhanced coverage arms were included in testing for antibodies (N = 24 communities) for logistical reasons.

Census

Trained study health workers conducted a full household census in all communities prior to the initial survey visit. During the baseline visit, adults in the household consented to census data collection, and study personnel recorded the name, sex, and age or date of birth for all individuals in the household.

Trachoma grading and conjunctival swab collection

After consent was obtained, study participants were examined for the presence of TF and TI. Clinical grading of each everted superior tarsal conjunctiva was performed using a 2.5x binocular loupe and a torch light, if necessary, per the WHO grading system. Clinical grading was performed according to the WHO simplified grading system of TF being the presence 5 or more follicles >0.5 mm in diameter and TI as inflammation severe enough to obscure 50% of deep tarsal vessels in one or both eyes [14]. Prior to swabbing, a trained photographer took at least 2 photographs of the right eyelid of all participants using a Nikon D-series camera and a Micro Nikon 105 mm; f/2.8 lens (Nikon, Tokyo, Japan). After conjunctival examination, a Dacron swab was passed 3 times over the right upper tarsal conjunctiva, rotating the swab approximately 120 degrees between each pass. All samples were placed immediately on cold packs in the field and transferred to -20°C within 10 hours, then shipped on cold packs to University of California, San Francisco, CA, USA where they were stored in -80°C freezers until processing.

PCR testing

PCR testing was performed for children aged 0-5 years. Samples from the same village, age in years, and visit were randomly pooled into groups of five for group testing, with a possible remainder pool of one to four samples [15]. Pooled samples were tested for the presence of *Ct* DNA using the Roche Amplicor qualitative PCR assay (Roche Molecular Systems, Indianapolis, IN, USA). Community prevalence was estimated from the pools as previously described [11,15].

Interventions

In communities randomized to annual treatment, study participants age 6 months of age and older received a directly observed dose of oral azithromycin (20 mg/kg up to a maximum dose of 1 g in adults). In biannually treated communities, only children up to 12 years of age were offered treatment. Children under 6 months of age in all communities were offered topical tetracycline ointment (1%) to be applied to both eyes twice a day for six weeks. Pregnant women in the annual arm and individuals allergic to macrolides were offered topical tetracycline. All communities were visited up to four days in order to achieve 90% treatment coverage [12,13].

Blood collection and antibody testing

Children under 5 years of age were selected randomly in each village for blood sample collection via finger stick or heel stick, with a goal of 50 children per village. Blood spots were analyzed for antibody to *Ct* antigens Pgp3 and CT694 using a multiplex bead array assay on a Luminex 200 platform, as previously described [7]. Results were reported as median fluorescence intensity minus background (MFI-BG) where background is the signal from beads run with buffer only. Positivity cut-off for Pgp3 was greater than or equal to 1083, and CT694 cutoff was greater than or equal to 496 as determined by receiver operator characteristic (ROC) curve analysis from a pediatric U.S.-based negative panel (N = 117) and Tanzania based positive panel from children with ocular Ct infection (N = 40) [7].

Statistical analysis

Data were entered into a customized database (Microsoft Access v2007) developed at the Dana Center, Johns Hopkins University. To estimate associations between seropositivity, clinical trachoma, and age at the individual level, we used generalized linear models with a binomial distribution and log link to estimate prevalence ratios (PR). All standard errors were clustered at the community level, which was the randomization unit of the study. As individual-level PCR data were not available, associations between seropositivity to the *Ct* antigen and ocular chlamydia infection were conducted only at the community level. We additionally analyzed the association between seropositivity and clinical trachoma at the cluster level. We used linear regression models to evaluate relationships between trachoma indicators at the community level. All analyses were conducted in Stata 14.1 (StataCorp, College Station, TX).

Ethics

All procedures and protocols for this study were approved by the Committee for Human Research of UCSF, and le Comité Consultatif National d'Ethique du Minstère de la Santé Publique, Niger (Ethical Committee, Niger Ministry of Health). The study's Data Safety and Monitoring Committee observed the study implementation during annual reviews of quality assurance, as appointed by the PRET study Executive Committee. All village leaders of communities within the study agreed to participate in the trial with written (thumbprint) consent. For children under the age of 16, consent was given by a parent or a guardian. All persons participating in the trial were given the opportunity to be treated according to their community's random treatment assignment. Communities not included in the study were offered treatment through the national treatment program. CDC personnel did not have access to personal identifying information and were determined to be non-engaged in the study.

Results

At the 36-month follow-up, 988 1–5-year-old children from 24 communities had clinical trachoma and serology data available. TF prevalence was 7.8% (95% CI 6.1 to 9.5%) and TI prevalence was 1.6% (95% CI 0.9 to 2.6%) (Table 1). The overall prevalence of antibody positivity to Pgp3 was 27.2% (95% CI 24.5 to 30%), and to CT694 was 23.7% (95% CI 21 to 26.2%). Community prevalence of ocular chlamydia in 0-5-year-olds was 5.2% (95% CI 2.8 to 7.6%).

Individual-level association of seropositivity with active trachoma

Children with antibodies to either Pgp3 or CT694 were more likely to have TF (PR, 1.90, 95% CI 1.49 to 2.43, P<0.001) and also TI, although the latter relationship was not statistically significant (PR 1.65, 95% CI 0.99 to 2.75, P = 0.06).

Community-level association of seropositivity with active trachoma and ocular chlamydia infection

Fig 1 shows the community-level prevalence of TF and ocular chlamydia compared to seropositivity to Pgp3 and/or CT694. At the community level, Pgp3 and/or CT694 seroprevalence was significantly correlated with ocular chlamydia infection (linear regression coefficient 0.19, 95% CI 0.08 to 0.29, P = 0.001), TF prevalence (linear regression coefficient 0.25, 95% CI 0.13 to

Table 1. Community and individual-level trachoma indicators, overall and by age group.

	N	TF+	TI+	Pgp3+	CT694+	PCR+ ¹
Community-level prevalence, (95% CI)	24	7.1% (4.2 to 10.0%)	1.8% (0.4 to 3.2%)	25.6% (17.4 to 33.8)	22.3% (16.7 to 27.9%)	5.2% (2.8 to 7.6%)
Individual-level prevalence, N (%)	988	77 (7.8%)	16 (1.6%)	269 (27.2%)	234 (23.7%)	
Age Subgroup (year) Individual-level, N, (%)						
1	212	12 (5.7%)	5 (2.4%)	31 (14.6%)	32 (15.1%)	
2	191	18 (9.4%)	3 (1.6%)	37 (19.4%)	32 (16.8%)	
3	173	14 (8.1%)	3 (1.7%)	41 (23.7%)	42 (24.3%)	
4	197	19 (9.6%)	3 (1.5%)	67 (34.0%)	49 (24.9%)	
5	215	14 (6.5%)	2 (0.9%)	93 (43.3%)	79 (36.7%)	
Community-level association with Pgp3/CT694 ² , mean difference (95% CI)	24	0.25 (0.13 to 0.36)	0.07 (0.004 to 0.14)			0.19 (0.08 to 0.29)
Individual-level association with Pgp3/CT694 ³ , prevalence ratio (95% CI)	988	1.90 (1.49 to 2.43)	1.65 (0.99 to 2.75)			

Abbreviations: TF, trachomatous inflammation-follicular; TI, trachomatous inflammation-intense; CI, confidence interval

¹Due to pooling, only community-level prevalence estimates are available for *C. trachomatis* infection as determined by PCR

²Linear regression model

³Generalized linear model with a binomial distribution and log link, with standard errors clustered at the community level

https://doi.org/10.1371/journal.pntd.0007127.t001

0.36, P<0.001), and TI prevalence (linear regression coefficient 0.07, 95% CI 0.004 to 0.14, P = 0.04).



Fig 1. Community level association between seropositivity to Pgp3 and/or CT694 and trachoma indicators. Fig 1A depicts the association between seropositivity to Pgp3 and/or CT694 and trachomatous inflammation–follicular (TF), with community-level seroprevalence on the Y-axis and community-level TF prevalence on the X-axis. Fig 1B depicts the association between seropositivity to Pgp3 and/or CT694 and ocular chlamydia infection, with community-level seroprevalence on the Y-axis and community-level ocular chlamydia prevalence on the X-axis. Circles represent individual communities, blue lines represent the linear regression line and grey shading represents 95% confidence intervals.

https://doi.org/10.1371/journal.pntd.0007127.g001

Age trends

The probability of having an antibody response to Pgp3 or CT694 increased with increasing age (PR 1.25 per one-year increase in age, 95% CI 1.17 to 1.35, *P*-trend<0.001; Fig 2). Older age was not significantly associated with a diagnosis of TF (PR 1.02 per year, 95% CI 0.89 to 1.16, *P*-trend = 0.82) or TI (PR 0.87, 95% CI 0.60 to 1.25, *P*-trend = 0.45).



Fig 2. Prevalence of antibodies recognizing Pgp3 (blue) and CT694 (red), trachomatous inflammation—follicular (TF, purple) and trachomatous inflammation—intense (TI, green) by age in years.

https://doi.org/10.1371/journal.pntd.0007127.g002

	Annual (N = 12)	Biannual (N = 12)
TF prevalence (95% CI)	8.9% (3.3 to 14.4%)	5.4% (2.9 to 7.9%)
TI prevalence (95% CI)	1.7% (0 to 3.9%)	1.9% (0 to 4.0%)
C. trachomatis prevalence (95% CI)	7.1% (2.7 to 11.4%)	3.3% (1.0 to 5.5%)
Seropositivity to Pgp3 (95% CI)	28.4% (16.5 to 40.3%)	22.8% (9.8 to 35.8%)
Seropositivity to CT694 (95% CI)	23.5% (14.3 to 32.6%)	21.0% (13.1 to 29.0%)

Table 2. Community-level trachoma indicators by study arm.

https://doi.org/10.1371/journal.pntd.0007127.t002

Annual vs biannual azithromycin distribution

There was no significant difference between study arms in the percentage of children antibody-positive to Pgp3 (PR 0.72 biannual versus annual, 95% CI 0.39 to 1.33, P = 0.29) or CT694 (PR 0.81 biannual versus annual, 95% CI 0.51 to 1.28, P = 0.36; Table 2).

Discussion

The presence of antibodies to Ct antigens was correlated with both TF and ocular Ct infection following a 36-month annual or biannual mass azithromycin distribution program in a trachoma-endemic region of Niger. However, there was no difference in serologic outcomes by study arm, consistent with clinical data suggesting that biannual treatment did not significantly alter transmission of ocular chlamydia compared to annual treatment [11]. That serologic outcomes were consistent with other trachoma indicators supports the finding that antibodies to Ct antigens are correlated with TF and ocular Ct infection and provide complementary information. Future work evaluating serologic outcomes in a trial with a significant effect on ocular Ct infection or TF would provide additional evidence about the relationship between these indicators.

Currently, decision-making for MDA in trachoma elimination programs relies solely on clinical grading of TF. Grading of clinical trachoma is subjective, and prevalence surveys have demonstrated that there is poor agreement between clinical disease and ocular Ct infection, particularly after multiple rounds of antibiotic treatment [2,3,14,16]. Additionally, TF may be observed in the absence of infection, either as residual inflammation from the etiologic agent Ct[17,18], or due to other non-chlamydial bacteria [19]. Point prevalence of TF, or a test of infection, reflects disease or infection state at a current point in time, whereas serologic patterns may allow for identification of longer-term patterns in Ct transmission [7,8]. Anti-Ct antibody responses increased with age in this study, whereas TF prevalence was not significantly different across age groups, suggesting that antibody positivity rates represent the pool of exposed individuals rather than currently or recently-infected ones.

While PCR assessment of ocular Ct infection and clinical grading for TF and TI represent cross-sectional prevalence of trachoma, age-seroprevalence curves may provide additional insight to changes in transmission of ocular Ct over time. PCR or NAAT testing has historically been too costly for use in program settings[20] but cost-effective PCR tests are now being evaluated in program contexts [6]. In ocular swab specimens, PCR tests for Ct infection are a more specific indicator for the causative agent of trachoma than antibody testing in sera, as antibody responses to the antigens we studied cannot differentiate between exposure to ocular Ct and other chlamydial infections. Perinatal transmission of Ct from mothers with urogenital chlamydia could potentially lead to seropositivity among young children, as could sexual exposure in individuals after the age of sexual debut. However, focusing on the younger children and on the age-seroprevalence curve rather than absolute rates of seropositivity may allow for distinction between ongoing ocular Ct transmission and a single exposure. For example, in the

Solomon Islands, a lack of increase in seropositivity to Pgp3 in 1–9-year-olds correlated with low infection rates, despite the 26% prevalence of TF, contrasting to the steep increase in sero-positivity with age observed amongst 1–9-year-olds in Kiribati where TF prevalence was 28% and infection prevalence was 24% [21,22].

Antibody testing for chlamydial antigens has been conducted in a number of trachoma program settings. In treatment-naïve communities, the slope of the age seroprevalence curve increased with increasing community TF prevalence [4,21]. A significant decline in antibody responses has been shown following mass azithromycin distribution compared to pre-treatment levels in a cross-sectional study [8]. In this mesoendemic region of Niger, we noted more than 30% prevalence of antibodies to Pgp3 and CT694, consistent with results from mesoendemic communities in Tanzania [4]. In settings where surveillance surveys for trachoma elimination have been conducted, age seroprevalence curves corresponded to decreases in trachoma transmission [23]. Seroprevalence of antibodies to Pgp3 in children ages 1 to 9 years in these surveys ranged from less than 2% in some surveys[24,25] to as high as 7.5%[26] but without a steep increase in age seroprevalence curves seen in settings with ongoing transmission. The current data add to this body of knowledge by evaluating antibody responses at impact surveys after 3 rounds of MDA, a program setting for which limited data exist.

TF prevalence in this study was 7.5% after 3 years of MDA and thus antibody data do not come from a setting in which elimination thresholds have been achieved. Furthermore, the communities included in this analysis were treated with an enhanced coverage target, with up to four days of treatment. Antibiotic coverage may have been higher than is seen in programmatic or trial settings with lower coverage targets. The antibody responses and curves therefore may not be representative of what would be seen in a previously-endemic setting that has reached the elimination threshold. Trachoma programs typically include children up to age 9 in monitoring. Other studies have shown further increases in antibody responses in children aged 6–9 years[21,26], and inclusion of those ages here may have improved our ability to draw inferences from the shape of the curve. Since ocular swabs were pooled for PCR analysis, we were unable to obtain individual-level correlations between PCR and antibody positivity.

WHO currently recommends use of TF for deciding programmatic endpoints. Laboratory testing, including *Ct* serology, could be used as a supplement or replacement for TF when conducting surveillance after validation of the elimination of trachoma as a public health problem, given serology is generally inexpensive, objective, and provides estimates of exposure over time.[26] The elimination thresholds do not require a complete absence of ocular *Ct* infection, and therefore infection may still be present in communities that have reached the district-wide elimination threshold, as was seen in Tanzania where infection was present but did not lead to re-emergence.[27] Having a test of exposure, or of repeated infection, would allow more complete evaluation of the history of exposure to ocular *Ct* in children.

The findings and conclusions in this report are those of the authors and do not necessarily represent the official position of the Centers for Disease Control and Prevention.

Supporting information

S1 Checklist. STROBE checklist for cross-sectional studies. (DOC)

Acknowledgments

Meeting Presentation: Results presented at the American Society of Tropical Medicine and Hygiene Annual Meeting, October 30, 2018.

Author Contributions

- **Conceptualization:** Jessica S. Kim, Catherine E. Oldenburg, Abdou Amza, Nicole E. Stoller, Sheila K. West, Robin L. Bailey, Jeremy D. Keenan, Bruce D. Gaynor, Thomas M. Lietman, Diana L. Martin.
- **Data curation:** Gretchen Cooley, Abdou Amza, Boubacar Kadri, Baido Nassirou, Sun Yu Cotter, Nicole E. Stoller, Sheila K. West, Robin L. Bailey, Jeremy D. Keenan, Bruce D. Gaynor, Travis C. Porco, Thomas M. Lietman, Diana L. Martin.
- Formal analysis: Jessica S. Kim, Catherine E. Oldenburg, Travis C. Porco, Thomas M. Lietman.
- **Funding acquisition:** Abdou Amza, Boubacar Kadri, Sun Yu Cotter, Sheila K. West, Robin L. Bailey, Bruce D. Gaynor, Thomas M. Lietman.
- **Investigation:** Gretchen Cooley, Abdou Amza, Baido Nassirou, Sun Yu Cotter, Nicole E. Stoller, Sheila K. West, Robin L. Bailey, Jeremy D. Keenan, Bruce D. Gaynor, Travis C. Porco, Thomas M. Lietman, Diana L. Martin.
- Methodology: Catherine E. Oldenburg, Gretchen Cooley, Boubacar Kadri, Baido Nassirou, Nicole E. Stoller, Sheila K. West, Robin L. Bailey, Jeremy D. Keenan, Bruce D. Gaynor, Travis C. Porco, Thomas M. Lietman, Diana L. Martin.
- Project administration: Abdou Amza, Boubacar Kadri, Baido Nassirou, Sun Yu Cotter, Sheila K. West, Robin L. Bailey, Jeremy D. Keenan, Bruce D. Gaynor, Travis C. Porco, Thomas M. Lietman.
- **Resources:** Abdou Amza, Sun Yu Cotter, Nicole E. Stoller, Sheila K. West, Robin L. Bailey, Bruce D. Gaynor, Travis C. Porco, Thomas M. Lietman.
- Software: Catherine E. Oldenburg, Travis C. Porco.
- **Supervision:** Abdou Amza, Boubacar Kadri, Baido Nassirou, Sun Yu Cotter, Nicole E. Stoller, Sheila K. West, Robin L. Bailey, Jeremy D. Keenan, Bruce D. Gaynor, Thomas M. Lietman, Diana L. Martin.
- Validation: Catherine E. Oldenburg.
- Visualization: Catherine E. Oldenburg.
- Writing original draft: Jessica S. Kim, Catherine E. Oldenburg, Diana L. Martin.
- Writing review & editing: Gretchen Cooley, Abdou Amza, Boubacar Kadri, Baido Nassirou, Sun Yu Cotter, Nicole E. Stoller, Sheila K. West, Robin L. Bailey, Jeremy D. Keenan, Bruce D. Gaynor, Travis C. Porco, Thomas M. Lietman.

References

- 1. World Health Organization. Report of the 17th Meeting of the WHO Alliance for the Global Elimination of Blinding Trachoma. Geneva: 2013.
- Ramadhani AM, Derrick T, Macleod D, Holland MJ, Burton MJ. The Relationship between Active Trachoma and Ocular Chlamydia trachomatis Infection before and after Mass Antibiotic Treatment. PLoS Negl Trop Dis 2016; 10:e0005080–26. https://doi.org/10.1371/journal.pntd.0005080 PMID: 27783678
- Zambrano AI, Munoz B, Mkocha H, et al. Measuring Trachomatous Inflammation-Intense (TI) When Prevalence Is Low Provides Data on Infection With Chlamydia trachomatis. Invest Ophthalmol Vis Sci 2017; 58:997–1000. https://doi.org/10.1167/iovs.16-20421 PMID: 28535271
- Martin DL, Wiegand R, Goodhew B, et al. Serological Measures of Trachoma Transmission Intensity. Sci Rep 2015;:1–5.

- Martin DL, Bid R, Sandi F, et al. Serology for Trachoma Surveillance after Cessation of Mass Drug Administration. PLoS Negl Trop Dis 2015; 9:e0003555–11. <u>https://doi.org/10.1371/journal.pntd.</u> 0003555 PMID: 25714363
- Butcher R, Houghton J, Derrick T, et al. Reduced-cost Chlamydia trachomatis-specific multiplex realtime PCR diagnostic assay evaluated for ocular swabs and use by trachoma research programmes. J Microbiol Methods 2017; 139:95–102. https://doi.org/10.1016/j.mimet.2017.04.010 PMID: 28487054
- Goodhew EB, Priest JW, Moss DM, et al. CT694 and pgp3 as Serological Tools for Monitoring Trachoma Programs. PLoS Negl Trop Dis 2012; 6:e1873–10. https://doi.org/10.1371/journal.pntd. 0001873 PMID: 23133684
- 8. Goodhew EB, Morgan SMG, Switzer AJ, et al. Longitudinal analysis of antibody responses to trachoma antigens before and after mass drug administration. 2014; 14:1–8.
- Wiegand RE, Cooley G, Goodhew B, et al. Latent class modeling to compare testing platforms for detection of antibodies against the Chlamydia trachomatis antigen Pgp3. Sci Rep 2018;:1–8. <u>https://doi.org/10.1038/s41598-017-17765-5</u>
- Gwyn S, Cooley G, Goodhew B, et al. Comparison of Platforms for Testing Antibody Responses against the Chlamydia trachomatis Antigen Pgp3. Am J Trop Med Hyg 2017; 97:1662–1668. <u>https://doi.org/10.4269/ajtmh.17-0292</u> PMID: 29016320
- Amza A, Kadri B, Nassirou B, et al. A cluster-randomized trial to assess the efficacy of targeting trachoma treatment to children. Clin Infect Dis 2017; 64:743–750. <u>https://doi.org/10.1093/cid/ciw810</u> PMID: 27956455
- Amza A, Kadri B, Nassirou B, et al. Effectiveness of expanding annual mass azithromycin distribution treatment coverage for trachoma in Niger: a cluster randomised trial. Br J Ophthalmol 2017;:bjophthalmol-2017-310916-8.
- Oldenburg CE, Amza A, Kadri B, et al. Comparison of Mass Azithromycin Coverage Targets of Children in Niger: A Cluster-Randomized Trachoma Trial. Am J Trop Med Hyg 2018; 98:389–395. https://doi. org/10.4269/ajtmh.17-0501 PMID: 29260659
- Thylefors B, Dawson CR, Jones BR, West SK, Taylor HR. A simple system for assessment of trachoma and its complications. Bull World Health Organ 1987;:477–483. PMID: 3500800
- Ray KJ, Zhou Z, Cevallos V, et al. Estimating Community Prevalence of Ocular Chlamydia trachomatis Infection using Pooled Polymerase Chain Reaction Testing. Ophthalmic Epidemiol 2014; 21:86–91. https://doi.org/10.3109/09286586.2014.884600 PMID: 24621121
- Munoz B, Stare D, Mkocha H, Gaydos C, Quinn T, West SK. Can Clinical Signs of Trachoma Be Used after Multiple Rounds of Mass Antibiotic Treatment to Indicate Infection? Invest Ophthalmol Vis Sci 2011; 52:8805–8806.
- Bailey R, Duong T, Carpenter R, Whittle H, Mabey D. The duration of human ocular Chlamydia trachomatis infection is age dependent. Epidemiol Infect 1999; 123:479–486. PMID: 10694161
- Keenan JD, Lakew T, Alemayehu W, et al. Clinical Activity and Polymerase Chain Reaction Evidence of Chlamydial Infection after Repeated Mass Antibiotic Treatments for Trachoma. Am J Trop Med Hyg 2010; 82:482–487. https://doi.org/10.4269/ajtmh.2010.09-0315 PMID: 20207878
- Burton MJ, Hu VH, Massae P, et al. What Is Causing Active Trachoma? The Role of Nonchlamydial Bacterial Pathogens in a Low Prevalence Setting. Invest Ophthalmol Vis Sci 2011; 52:6012–6016. https://doi.org/10.1167/iovs.11-7326 PMID: 21693601
- Harding-Esch E, Jofre-Bonet M, Dhanjal JK, et al. Costs of Testing for Ocular Chlamydia trachomatis Infection Compared to Mass Drug Administration for Trachoma in The Gambia: Application of Results from the PRET Study. PLoS Negl Trop Dis 2015; 9:e0003670–18. <u>https://doi.org/10.1371/journal.pntd.</u> 0003670 PMID: 25901349
- Cama A, Müller A, Taoaba R, et al. Prevalence of signs of trachoma, ocular Chlamydia trachomatis infection and antibodies to Pgp3 in residents of Kiritimati Island, Kiribati. PLoS Negl Trop Dis 2017; 11: e0005863–15. https://doi.org/10.1371/journal.pntd.0005863 PMID: 28898240
- Butcher R, Sokana O, Jack K, et al. Clinical signs of trachoma are prevalent among Solomon Islanders who have no persistent markers of prior infection with Chlamydia trachomatis. Wellcome Open Res 2018; 3:14. https://doi.org/10.12688/wellcomeopenres.13423.2 PMID: 29588922
- Migchelsen SJ, Iveda NS x000FA, Martin DL, et al. Serology reflects a decline in the prevalence of trachoma in two regions of The Gambia. Sci Rep 2017;:1–10. https://doi.org/10.1038/s41598-016-0028-x
- Sun MJ, Zambrano AI, Dize L, et al. Evaluation of a field test for antibodies against Chlamydia trachomatis during trachoma surveillance in Nepal. Diagn Microbiol Infect Dis 2017; 88:3–6. https://doi.org/ 10.1016/j.diagmicrobio.2017.01.004 PMID: 28214223

- West SK, Zambrano AI, Sharma S, et al. Surveillance surveys for reemergent trachoma in formerly endemic districts in Nepal from 2 to 10 years after mass drug administration cessation. JAMA Ophthalmol 2017; 135:1141–1146. https://doi.org/10.1001/jamaophthalmol.2017.3062 PMID: 28973295
- 26. West SK, Munoz B, Kaur H, et al. Longitudinal change in the serology of antibodies to Chlamydia trachomatis pgp3 in children residing in a trachoma area. Sci Rep 2018;:1–8. https://doi.org/10.1038/ s41598-017-17765-5
- West SK, Munoz B, Weaver J, et al. Can We Use Antibodies to Chlamydia trachomatis as a Surveillance Tool for National Trachoma Control Programs? Results from a District Survey. PLoS Negl Trop Dis 2016; 10:e0004352–11. https://doi.org/10.1371/journal.pntd.0004352 PMID: 26771906